Author’s response to reviews

Title: Acute Biliary Events During Anti-Tuberculosis Treatment: Hospital Case Series and Nationwide Cohort Study

Authors:

LIH-YU CHANG (pagasus2751@yahoo.com.tw)
Chih-Hsin Lee (chlee.tw@gmail.com)
Chia-Hao Chang (b89402012@ntu.edu.tw)
Ming-Chia Lee (ymkbaz60@gmail.com)
Meng-Rui Lee (sheepman1024@gmail.com)
Jann-Yuan Wang (jywang@ntu.edu.tw)
Li-Na Lee (linalee@ntu.edu.tw)

Version: 3 Date: 17 Dec 2017

Author’s response to reviews:

Responses to Editor

1. We note that the manuscript contains potentially identifiable information. It is BioMed Central policy to not publish more than 2 indirect identifiers without explicit consent for publication from the participants, as per this paper. Currently, the age and gender information, in combination with other identifiers, may compromise patient/participant anonymity. Please amend the information to address this issue, by removing this information or by providing ages as age-ranges.

Answer: Thanks for the suggestions. We have provided age-ranges instead of ages.

2. Please have the text edited by a professional language editing service or a native English speaking colleague. There are several issues with grammar, wording, spelling, and/or punctuation that need to be addressed before acceptance for publication.

Answer: Prior to this submission, we have let our manuscript edited by a professional language editor in the Wallace Academic Editing. Thank you.
3. Please add a ‘Declarations’ heading.

Answer: Thank you. We have added the ‘Declarations’ heading accordingly.

4. Thank you for including the ethical statement. Please duplicate this entire statement (including where consent was waived) within the Declarations section under the title ‘Ethics approval and consent to participate’.

Answer: We have revised our manuscript accordingly. Thank you.

5. We note that you have not included a ‘Consent for publication’ section in the Declarations. Consent for publication refers to consent for the publication of identifying images or other personal or clinical details of participants that compromise anonymity. Seeing as this is not applicable to your manuscript please state “Not Applicable” in this section.

Answer: Thank you for the instruction. This section has been revised accordingly.

6. Please provide figure titles/legends under a separate heading of 'Figure Legends' after the References.

Answer: Thank you for the reminders. We have provided figure legends under a separate heading of ‘Figure Legends’.

7. Please reformat your abstract so that it follows the following structure:

   Background: the context and purpose of the study

   Methods: how the study was performed and statistical tests used

   Results: the main findings

   Conclusions: brief summary and potential implications

Answer: Thanks for the correction. We have reformatted our abstract according to the journal style. Thank you.
8. Please include the email addresses for all authors on the title page. The corresponding author should still be indicated. Please also ensure all email addresses provided in the manuscript and the submission system match.

Answer: We have added the E-mail address for each author on the title page. Thank you.

9. Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

Answer: Thank you for the reminder. All authors have read and approved the manuscript, and we have mentioned about this in the Authors' contributions section. Thank you.

10. Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

Answer: Thank you for the instruction. We have added the abbreviation list before the Declaration section.

11. Please change the heading ‘Data sharing’ to ‘Availability of data and material’

Answer: Thank you. The heading has been revised accordingly.

12. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Answer: We have revised our manuscript and prepared a clean and final version. Thank you.